Author's response to reviews

Title: Radiological features do not predict failure of two-stage arthroplasty for prosthetic joint infection: a retrospective case-control study

Authors:

Susanna Dunachie (susie.dunachie@ndm.ox.ac.uk)
James Teh (james.teh@ouh.nhs.uk)
Vivian Ejindu (vivian.ejindu@stgeorges.nhs.uk)
Philip Bejon (pbejon@well.ox.ac.uk)
Hemant Pandit (hemant.pandit@ouh.nhs.uk)
Ivor Byren (ivor.byren@ouh.nhs.uk)

Version: 2 Date: 13 September 2013

Author's response to reviews:

Dear editorial team

Thank you for your email requesting a revision to the manuscript to include more details about the ethics statement. This section of the manuscript (p3, Methods, study population) now reads:

"There was no research-related contact with patients. In response to our enquiry, our institutional review board (Oxfordshire Research Ethics Committee) advised informed consent and ethical approval was not required. All activity was conducted in accordance with the Declaration of Helsinki, and national and institutional standards."

Please let me know if you require any further information.

Best wishes,

Dr Susanna Dunachie